Author's response to reviews

Title: Comprehensive Expressional Analyses of Antisense Transcripts in Colon Cancer Tissues Using Artificial Antisense Probes

Authors:

Rintaro Saito (rsaito@sfc.keio.ac.jp)
Keisuke Kohno (c0730177@md.tsukuba.ac.jp)
Yuki Okada (okayan@gmail.com)
Yuko Osada (yuko.osada@gmail.com)
Koji Numata (numata@brc.riken.jp)
Chihiro Kohama (s0921075@u.tsukuba.ac.jp)
Kazufumi Watanabe (watanabe@hssnet.co.jp)
Hajime Nakaoka (nakaoka@cslab.co.jp)
Naoyuki Yamamoto (yamamoto@cslab.co.jp)
Akio Kanai (akio@sfc.keio.ac.jp)
Hiroshi Yasue (hyasue@affrc.go.jp)
Soichiro Murata (soichiro@md.tsukuba.ac.jp)
Kuniya Abe (abe@rtc.riken.jp)
Masaru Tomita (mt@sfc.keio.ac.jp)
Nobuhiro Ohkohchi (nokochi3@md.tsukuba.ac.jp)
Hidenori Kiyosawa (hkiyosaw@lab.nig.ac.jp)

Version: 4 Date: 28 February 2011

Author's response to reviews: see over
Response to referee’s comment

1. Reviewer #1 (Version: 2 Date: 29 November 2010 Reviewer: Andreas Werner)

Comment 1·1: There is one little detail that the authors should clarify: To me, the sentence "These results are consistent with pervious reports suggesting that these RNAs are possibly associated with transcription." needs specification (what aspect of transcription).

Response 1·1: We clarified the sentence with addition of relevant citation (Page 18, marked in yellow).